Author's response to reviews

**Title:** Can biting of the tongue kill a patient? Acquired hemophilia as cause of life threatening hemorrhage in an elderly patient: A case report

**Authors:**

Theodoros Kelesidis (tkelesid@gmail.com)  
Jonelle Raphael (rjonell@cchs.org)  
Elizabeth Blanchard (bezizab@cchs.org)  
Rekha Parameswaran (pararek@cchs.org)

**Version:** 3  **Date:** 8 May 2010

**Author's response to reviews:**

Response to Reviewers’s comments

Major revisions (we require the authors to make these changes)

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

This was addressed in the revised manuscript

Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

This was addressed in the revised manuscript

Abstract: please ensure that the abstract is identical in the manuscript and the submission system.

This was addressed in the revised manuscript

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Manuscript format: please remove all review tracking.

This was addressed in the revised manuscript

Abstract: please remove the keywords section.

This was addressed in the revised manuscript

Figures in text: please remove figure 1 from the main manuscript file as it has been separately uploaded in the submission system.

This was addressed in the revised manuscript